Author's response to reviews

Title: Extending life for people with a terminal illness: a moral right or an expensive death? Exploring societal perspectives

Authors:

Neil McHugh (neil.mchugh@gcu.ac.uk)
Rachel Baker (rachel.baker@gcu.ac.uk)
Helen Mason (helen.mason@gcu.ac.uk)
Laura Williamson (laura.williamson@gcu.ac.uk)
Job van Exel (vanexel@bmg.eur.nl)
Rohan Deogaonkar (rohan.deogaonkar@gcu.ac.uk)
Marissa Collins (marissa.collins@gcu.ac.uk)
Cam Donaldson (cam.donaldson@gcu.ac.uk)

Version: 2
Date: 22 September 2014

Author's response to reviews:

Response to Reviewers

RE: Extending life for people with a terminal illness: a moral right or an expensive death? Exploring societal perspectives

1. Line and page numbers have been added.
2. The authors declare that they have no competing interests in the ‘Competing Interests’ section.
3. An ‘Authors’ Contribution’ section has been added.